Author's response to reviews

Title: Health-related quality of life (HRQOL) in children and adolescents as an outcome criterion to evaluate family oriented support for young carers in Germany: a review of the literature

Authors:

Jörg große Schlarmann (schlarmann@uni-wh.de)
Sabine Metzing-Blau (smetzing@uni-wh.de)
Wilfried Schnepp (schnapp@uni-wh.de)

Version: 2 Date: 21 August 2008

Author's response to reviews:

We made the following modifications to our manuscript:
1) Title: changed "a review of the literature" to "an integrative review of the literature"
2) Abstract: modified the Method section
3) Methods: added a flow-chart of our search strategy.